Author's response to reviews

Title: The Aftermath of Adverse Events in Spanish Primary Care and Hospital Health Professionals

Authors:

José Joaquín Mira (jose.mira@umh.es)
Irene Carrillo (icarrillo@umh.es)
Susana Lorenzo (slorenzo@fhalcorcon.es)
Lena Ferrús (lena.ferrus@sanitatintegral.org)
Carmen Silvestre (CARMEN.SILVESTREBUSTO@osakidetza.net)
Pastora Pérez-Pérez (pastora.perez.perez@juntadeandalucia.es)
Guadalupe Olivera (guadalupe.olivera@salud.madrid.org)
Fuencisla Iglesias (figlesias@jccm.es)
Elena Zavala (ELENA.ZAVALAAIZPURUA@OSAKIDETZA.NET)
José Angel Maderuelo-Fernández (jmaderuelo@saludcastillayleon.es)
Julián Vitaller (vitaller_jul@gva.es)
Roberto Nuño-Solinís (nuno@bioef.org)
Pilar Astier (mpastier@gmail.com)

Version: 2 Date: 30 September 2014

Author's response to reviews:

Thanks for your kindly message. We have tried to introduce the changed required:
1. Authors’ emails are now included
2. This version includes line and page numbers
3. All sections are now included
4. Tables have been moved

Yours sincerely